Author’s response to reviews

Title: Protocol for a randomised controlled trial of treatment of asymptomatic candidiasis for the prevention of preterm birth [ACTRN12610000607077]

Authors:

Christine L Roberts (clroberts@med.usyd.edu.au)
Jonathan M Morris (jmorris@med.usyd.edu.au)
Kristen R Rickard (KRRickard@nsccahs.health.nsw.gov.au)
Warwick B Giles (wgiles@nsccahs.health.nsw.gov.au)
Judy M Simpson (judys@health.usyd.edu.au)
George Kotsiou (gkotsiou@med.usyd.edu.au)
Jennifer R Bowen (JBowen@nsccahs.health.nsw.gov.au)

Version: 2 Date: 9 December 2010

Author’s response to reviews:

Dear Editor

BMC Pregnancy and Childbirth

Further to our correspondence with Nina Titmus we are submitting a revised version of our trial protocol. In the revised version we have removed repetition (with our pilot study manuscript, MS: 1280202917436949) pages 9-12 and have added a statement of context for the pilot study (page 11). Please note that we have cited each paper (the pilot study and protocol) in the other in putting them into context, even though at this stage both are unpublished.

Thank you for your attention and we look forward to hearing from you.

Yours sincerely

Christine Roberts